Author's response to reviews

Title: Prevalence and distribution of Actinobacillus actinomycetemcomitans and its cdt gene in subgingival plaque of Chinese periodontitis patients

Authors:

Xiaoqian Wang (xiaoqian0103@hotmail.com)
Lu Li (lelunia@foxmail.com)
Mifang Yang (yangmifang@qq.com)
Yng Geng (839926010@qq.com)
Huiping Chen (18262631755@139.com)
Yan Xu (yanxu@njmu.edu.cn)
Ying Sun (ebolasun@njmu.edu.cn)

Version: 2 Date: 26 January 2014

Author's response to reviews: see over
1. We listed all the authors’ email address of co-authors on title page.
2. We listed all author's contribution.
3. We put the ‘Competing interests’ section after the Conclusions.
4. We put the Acknowledgements section between the Authors’ contributions section and Reference list.